\begin{table}%t4
%%\centering

\par
\caption{Outflows and associated sources in T Tauri.\label{tab:star+flow}}
\begin{tabular}{ccccc}
\hline \hline \noalign{\smallskip}
Star & Outflow & H$_2$ flow & PA($^{\circ}$) & Inclination ($^{\circ}$)\\
\hline
N & E -- W & 5,6, ``bow'' & $235{-}245$\tablefootmark{a},\tablefootmark{b} & 19--23\tablefootmark{c}, \tablefootmark{d}\\
Sa & SE -- NW & 1a/1b, 2 & 100--190\tablefootmark{a} & 79\tablefootmark{e}\\
Sb & SW & 3 & 210--250\tablefootmark{a},\tablefootmark{b} & 35--60\tablefootmark{a}, \tablefootmark{g}\\
\hline
\end{tabular}
\tablefoot{\tablefoottext{a}{This work;} \tablefoottext{b}{\cite{loinard2007};} \tablefoottext{c}{\cite{herbst1997};} \tablefoottext{d}{\cite{eisloffel1998};} \tablefoottext{e}{\cite{solf1999};}  \tablefoottext{g}{\cite{ratzka2009}}.}
\vspace*{-3mm}
\end{table}